Author's response to reviews

Title: Democracy: the forgotten challenge for bioethics in the developing countries

Authors:

Ghaiath M. A. Hussein (ghaiathme@gmail.com)

Version: 6 Date: 18 May 2009

Author's response to reviews: see over
Dear Editor,

Thank you for the acceptance of my manuscript for publication in your journal. I have reviewed the above manuscript according to the formatting changes requested and made the necessary changes. They are as follows:

-----------------------------
Major revisions (We require the author to make these changes)

Author list - Please remove the job titles from the title page (e.g Dr / Professor). You may wish to include an About the Authors' section after the authors' contributions.

Any titles were removed

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

Manuscript sections - Manuscript sections were divided (in the following order):

<table>
<thead>
<tr>
<th>Section</th>
<th>Page</th>
</tr>
</thead>
<tbody>
<tr>
<td>Abstract</td>
<td>1</td>
</tr>
<tr>
<td>Background</td>
<td>2</td>
</tr>
<tr>
<td>Discussion</td>
<td>5</td>
</tr>
<tr>
<td>Summary (named conclusion)</td>
<td>6</td>
</tr>
<tr>
<td>Abbreviations (if any)</td>
<td>Not applicable (N/A)</td>
</tr>
</tbody>
</table>
Authors' contributions - please use the following format (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

The section was added [page 8] and now reads: ‘GH has developed the argument, reviewed the literature and prepared the whole manuscript for submission’.

Reference 2, In press - Has this article now been published? If so, please provide the citation details.

Reference 3, In press - Has this article now been published? If so, please provide the citation details.

Both references were corrected, and now read as follow:

Ref Type: Electronic Citation

Ref Type: Magazine Article

References * Please provide full details for all references. This means for an Article providing volume/issue/page numbers and for a book the publisher/city. You can view the BMC reference style at the following link
http://www.biomedcentral.com/bmcgenet/ifora/#references
References are automatically formatted using Reference Manager (version 11). However, I could not find BMC Genetics style among the installed styles in Reference Manager.

References - Please place the reference before the full stop. For example `text [1].`
All references are placed between square brackets and placed before the full stop.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns, and no full stops/periods). For example you would change `The Title of My Prestigious Publication: A Scientific Study.`. To `The title of my prestigious publication: a scientific study`.
Capitalization was removed, and now the title reads as follows:

Democracy: the forgotten challenge for bioethics in the developing countries

Abstract - Abstracts must be structured into Background, Discussion, Summary. Please remember to also update the Abstract details on the submission page.
The abstract was re-structured as advised.

Please remove any blank pages.
No blank pages exist.

Please present all text in black
All text was changed into black

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.
Typographical and grammatical errors have been reviewed and corrected to the best possible.